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NuMI-MINOS Status Report  
Carlos Escobar/UNICAMP

Covering  week: 09/20 to 09/27

NuMI beam off since September 17 at 07:00h.

And will remain off until new target (NT05) is ready to be 
installed: current estimate is first week of November.
All MINOS shifts are cancelled until then.

On the positive side:

ACNET beam data logger problems: SOLVED (thank 
you to all involved)
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 Summary                      

• ND Summary
On all the period: 99.7% live time

• FD Summary
problems with one  Read Out Processor
made DAQ crash over the weekend, not possible 
to recover despite intervention by the mine crew 
on­call.
Replace it today (09/27)
FD livetime: 82%
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